\begin{thebibliography}{10}

\bibitem{AcharyaS96Study}
A.~Acharya and J.~Saltz.
\newblock {A Study of Internet Round-Trip Delay}.
\newblock Technical Report CS-TR-3736, University of Maryland, December 1996.

\bibitem{AcharyaS97Dynamic}
A.~Acharya and J.~Saltz.
\newblock {\em {Dynamic Linking for Mobile Programs}}, pages 245--62.
\newblock Springer Verlag, 1997.
\newblock {\em Mobile Object Systems}, Jan Vitek and Christian Tschudin (eds).

\bibitem{ACHARYA96}
A.~Acharya, M.~Uysal, R.~Bennett, A.~Mendelson, M.~Beynon, J.~K. Hollingsworth,
  J.~Saltz, and A.~Sussman.
\newblock Tuning the performance of {I/O} intensive parallel applications.
\newblock In {\em Proceedings of the Fourth Annual Workshop on I/O in Parallel
  and Distributed Systems (IOPADS)}, May 1996.

\bibitem{AdlTabatabaiLLW96}
A.~Adl-Tabatabai, G.~Langdale, S.~Lucco, and R.~Wahbe.
\newblock {Efficient and Language-Independent Mobile Programs}.
\newblock In {\em Proceedings of the SIGPLAN'96 Conference on Programming
  Language Design and Implementation}, pages 127--36, May 1996.

\bibitem{AmsalegFT96}
L.~Amsaleg, M.~Franklin, and A.~Tomasic.
\newblock {Query Scrambling for Bursty Data Arrival}.
\newblock Technical Report CS-TR-3714, Dept of Computer Science, University of
  Maryland, 1996.

\bibitem{AmsalegFTU96}
L.~Amsaleg, M.~Franklin, A.~Tomasic, and T.~Urhan.
\newblock {Scrambling Query Plans to Cope With Unexpected Delays}.
\newblock In {\em Proceedings of the Fourth International Conference on
  Parallel and Distributed Information Systems}, December 1996.

\bibitem{Borenstein94}
N.~Borenstein.
\newblock {Email With a Mind of its Own: The Safe-TCL Language for Enabled
  Mail}.
\newblock In {\em Proceedings of IFIP Working Group 6.5 International
  Conference}, pages 389--402, Jun 1994.

\bibitem{BadrinathW95}
{B.R.Badrinath and Girish Welling}.
\newblock {Event Delivery Abstractions for Mobile Computing}.
\newblock Technical Report LCSR-TR-242, {Rutgers University}, 1996.

\bibitem{NoblePS95}
{Brian D. Noble and Morgan Price and M.Satyanarayanan}.
\newblock {A Programming Interface for Application-Aware Adaptation in Mobile
  Computing}.
\newblock {\em {{Proceedings of the Second USENIX Symposium on Mobile and
  Location Independent Computing}}}, Feb. 1995.

\bibitem{Cardelli95}
L.~Cardelli.
\newblock {A Language With Distributed Scope}.
\newblock In {\em Proceedings of the 22nd ACM Symposium on Principles of
  Programming Languages}, Jan. 1995.

\bibitem{ClamenLNW90}
S.~Clamen, L.~Leibengood, S.~Nettles, and J.~Wing.
\newblock {Reliable Distributed Computing with Avalon/Common Lisp}.
\newblock In {\em {Proceedings of the International Conference on Computer
  Languages}}, pages {169--79}, 1990.

\bibitem{Falcone87}
J.~Falcone.
\newblock {A Programmable Interface Language for Heterogeneous Systems}.
\newblock {\em {ACM Transactions on Computer Systems}}, 5(4):{330--51}, Nov.
  1987.

\bibitem{GoslingM95}
J.~Gosling and H.~McGilton.
\newblock {The Java Language Environment White Paper}, 1995.

\bibitem{Gray96}
R.~Gray.
\newblock {Agent TCL: A Flexible and Secure Mobile-agent System}.
\newblock In {\em {Proceedings of the Fourth Annual Tcl/Tk Workshop (TCL 96)}},
  July 1996.

\bibitem{JohansenRS95}
D.~Johansen, R.~van Renesse, and F.~Schneider.
\newblock {An Introduction to the {TACOMA} Distributed System Version 1.0}.
\newblock Technical Report {95-23}, {University of Tromso}, 1995.

\bibitem{JulLHB88}
E.~Jul, H.~Levy, N.~Hutchinson, and A.~Black.
\newblock {Fine-Grained Mobility in the Emerald System}.
\newblock {\em {ACM Transactions on Computer Systems}}, 6(2):{109--33}, Feb.
  1988.

\bibitem{StamosG90}
{J.W. Stamos and D.K. Gifford}.
\newblock {Implementing Remote Evaluation}.
\newblock {\em {{IEEE Transactions on Software Engineering}}},
  16(7):{710--722}, July 1990.

\bibitem{Katz96}
R.~Katz.
\newblock {The Case for Wireless Overlay Networks}.
\newblock Invited talk at the ACM Federated Computer Science Research
  Conferences, Philadelphia, 1996.

\bibitem{eos}
M.~King.
\newblock {Earth Observation System Project Science} homepage.
\newblock {\em http://\-spso.gsfc.nasa.gov/\-spso\_homepage.html}, 1995.

\bibitem{LangeO96}
D.~Lange and M.~Oshima.
\newblock {\em {Programming Mobile Agents in Java}}.
\newblock In progress, 1996.
\newblock (ch 2,3).

\bibitem{MummertES95}
L.~Mummert, M.~Ebling, and M.~Satyanarayanan.
\newblock {Exploiting Weak Connectivity for Mobile File Access}.
\newblock In {\em {Proceedings of the Fifteenth ACM Symposium on Operating
  System Principles}}, Dec. {1995}.

\bibitem{StraberBH96}
M.~Stra{$\beta$}er, J.~Baumann, and F.~Hohl.
\newblock {Mole - A Java Based Mobile Agent System}.
\newblock In {\em Proceedings of the ECOOP'96 workshop on Mobile Object
  Systems}, 1996.

\bibitem{JavaRMI}
{Sun Microsystems}.
\newblock {\em Java Remote Method Invocation}.
\newblock {\em http://chatsubo.javasoft.com/\-current/\-rmi/\-index.html}.

\bibitem{White96}
J.~White.
\newblock {Telescript Technology: Mobile Agents}, 1996.
\newblock {\em http://www.genmagic.com\-/Telescript\-/Whitepapers}.

\end{thebibliography}

\section{Introduction\label{intro}}


The Internet is characterized by large fluctuations in resource
availability and quality.  These variations are primarily caused by
congestion and as such their occurrence cannot be predicted
reliably. The magnitude of the problem is illustrated by
Figure~\ref{fig:bw-var} which shows both short-term and long-term
variation in end-to-end network bandwidth between two machines, one at
UCLA and the other at the University of Wisconsin. This is apparent to
anyone who uses a web browser over the period of a single day.

\begin{figure}[tp]
\begin{tabular}{cc}
\psfig{file=short-term-bandwidth.ps,width=3in}&
\psfig{file=long-term-bandwidth.ps,width=3in}\\
(a) Short-term variation &
(b) Long-term variation 
\end{tabular}
\caption{Variation in application-level network bandwidth.}
\label{fig:bw-var}
\end{figure}

One of the biggest challenges for efficient data access over the
Internet is effective adaptation to resource fluctuations.  There are
a variety of emerging high end applications that are particularly
likely to benefit from an ability to adapt to changing patterns of
network and computational resources. The applications fall into two
roughly defined categories, metacomputing applications and customized
multicast (customcast) applications. Metacomputing applications
require synthesis of information obtained from multiple sources.

Metacomputing applications may generate data products that combine
results obtained from a variety of data sources. Data sources can
consist of archived information, information generated by physical
simulations, information from scientific instruments such as light or
electron microscopes or data from sensors based on land, aircraft, or
satellites.

In metacomputing scenarios, we generate a sequence of data products
by using a number of programs that are configured as a pipeline. For
instance in defense and disaster response scenarios, there is a need
to generate data products that combine information from combinations
of satellite sensors, aircraft sensors and ground based sensors. These
synthesized data products need to be generated on an ongoing basis.

We anticipate that the process of data product generation will require
moving large amounts of data through variable quality network
links. There are often significant computational costs associated with
data combining.  Resource aware scheduling may have to occur due to
changes in network characteristics, changes in the computational
demands associated with data combination, or changes in availability
of computational resources.

As an example, suppose a major earthquake strikes and the decision is
made to provide massive emergency relief.  Crisis managers will need
to quickly combine archival, reconnaissance and seismic data to
delineate areas of maximal structural damage and merge this
information with census and business data to determine local resources
that could be mobilized. For the prototype that is modeling these
situations, dynamic hierarchies of mobile programs could be configured
into a pipeline to process and combine data from multiple sensors and
databases.

Significant characteristics of metacomputing applications are that (1)
the applications are data-intensive and coordinate data from
geographically distributed data sources, (2) while components must
interact with each other, they may use different types of parallelism
and different data formats, or execute only on specific platforms.

Customized multicast (customcast) applications involve customized
propagation of data obtained from a single source. A simple example of
customcast from the non-high performance computing domain involves
periodic distribution of news clippings selected from a database on
the basis of user profiles. High performance computing issues come
into play when information to be disseminated involves large
quantities of filtered sensor data. Customcast can play an important
role in complex metacomputing scenarios. Metacomputing applications
typically involve synthesizing data obtained from a variety of
sources. For instance, customcast would be used when several
metacomputing applications need to use different subsets of a given
data source.

An example of customcast arises in an application under development by
our group in the Department of Pathology at Johns Hopkins Medical
School.  The application (the Virtual Microscope) under development at
Johns Hopkins involves development of software that makes it possible
to carry out a realistic digital emulation of a high power light
microscope. Raw data is captured by scanning collections of full
microscope slides under high power. For each microscope slide, high
power images are captured at multiple focal planes. Once the image is
captured, this application will make it possible to emulate the
behavior of a microscope by allowing users to continuously move the
stage and to simulate changing magnification and focus. The Virtual
Microscope is being designed so that it will be possible to achieve an
interactive level of response as the same dataset is simultaneously
explored by multiple users.  Customcast can be employed due to spatial
and temporal locality. While each datasets is quite large (10-100
Gbytes per specimen), most users will access only the portions of the
data set that contain features that are of medical interest.
Moreover, due to psychophysical limitations of human data processing,
users will frequently dwell on portions of the dataset that are of
particular interest.


Mobile programs can move an active thread of control from one site to
another during execution. This flexibility has many potential
advantages.  For example, a program that searches distributed data
repositories can improve its performance by migrating to the
repositories and performing the search on-site instead of fetching all
the data to its current location. Similarly, an Internet
video-conferencing application can minimize overall response time by
positioning its server based on the location of its
users. Applications running on mobile platforms can react to a drop in
network bandwidth by moving network-intensive computations to a proxy
host on the static network. The primary advantage of mobility in these
scenarios is that it can be used as a tool to adapt to variations in
the operating environment.  Applications can use online information
about their operating environment and knowledge of their own resource
requirements to make judicious decisions about placement of
computation and data.

Many systems provide some form of support for program mobility.  The
simplest form of support is the ability to download code and execute
it to completion at a single site -- as provided by systems like
Omniware~\cite{AdlTabatabaiLLW96}, Safe-TCL~\cite{Borenstein94},
Java~\cite{GoslingM95}. Other systems like Avalon~\cite{ClamenLNW90},
NCL~\cite{Falcone87} REV~\cite{StamosG90} and Obliq~\cite{Cardelli95}
allow programs in execution to initiate computation on remote nodes
and wait for their completion. The most sophisticated support is
provided by systems like Agent TCL~\cite{Gray96},
Emerald~\cite{JulLHB88}, Mole~\cite{StraberBH96},
Aglets~\cite{LangeO96}, TACOMA~\cite{JohansenRS95} and
Telescript~\cite{White96} which permit an executing program to move
while it is in execution.

Programs that use mobility as a mechanism to adapt to resource changes
have three requirements that are not shared with other mobile
programs. First, they need to be aware of their execution environment.
In particular, they need to monitor the level and quality of resources
in their operating environment. We refer to this as the {\em
awareness} requirement. Second, they need to be able to react to
changes in resource availability. We refer to this as the {\em
agility} requirement. Third, they need to be able to control the way
in which resources are used on their behalf (by libraries and other
support code). We refer to this as the {\em authority} requirement.

In this paper, we describe the design and implementation of Sumatra,
an extension of Java that supports resource-aware mobile programs. We
also describe the design and implementation of a distributed resource
monitor that provides the information required by Sumatra programs.

We first discuss the constraints that the requirements of awareness,
agility and authority place on languages for mobile programs. We then
describe Sumatra and discuss how its design decisions have been guided
by these constraints. Next, we discuss the considerations that guide
the design of resource monitors, in particular, fault-tolerance and
the need to allow applications to control the use of resources on
their behalf.  We describe Komodo, a distributed resource monitor and
discuss how well its design meets the requirements. Both Sumatra and
Komodo have been implemented and are available to individuals and
organizations with a Sun JDK license. We provide implementation
details wherever appropriate. Finally, we describe our initial
experience with a prototypical application that combines and composes
weather images from multiple geographically-distributed sources.

In this paper, we assume that the reader is familiar with mobile code
languages. For introduction to mobile code languages and mobility
mechanisms, see
\cite{AdlTabatabaiLLW96,Borenstein94,Cardelli95,ClamenLNW90,Falcone87,GoslingM95,Gray96,JohansenRS95,JulLHB88,LangeO96,StamosG90,StraberBH96,White96}.


